Author's response to reviews

Title: Socio-economic and demographic Determinants of Under-five Mortality in rural northern Ghana

Authors:

    Edmund W Kanmiki (eddy100@hotmail.com)
    Ayaga A Bawah (aab2161@cumc.columbia.edu)
    Isaiah Agorinya (Isaiah.Agorinya@navrongo-hrc.org)
    Fabian S Achana (fsachana@gmail.com)
    John Koku Awoonor-williams (kawoonor@gmail.com)
    Abraham R Oduro (aroduro@yahoo.com)
    James Phillips (jfp2113@columbia.edu)
    James Akazil (akazjames@yahoo.com)

Version: 2 Date: 21 May 2014

Author's response to reviews:

Dear Sir,

I have accordingly inserted line numbers and removed the figures from the main manuscript. the figures have now been uploaded separately as instructed.

Best Regards

Edmund W. Kanmiki